Author’s response to reviews

Title: Advanced practice nurses’ and general practitioners’ first experiences with introducing the advanced practice nurse role to Swiss primary care: a qualitative study

Authors:
Stefan Gysin (stefan.gysin@iham-cc.ch;stefangysin@hotmail.com)
Beat Sottas (sottas@formative-works.ch)
Muriel Odermatt (muriel.odermatt@stud.unilu.ch)
Stefan Essig (stefan.essig@iham-cc.ch)

Version: 2 Date: 14 Nov 2019

Author’s response to reviews:

Dear Ms. Amanpreet Athwal

Thank you for the assessment of our submission. We revised our manuscript according to your comments. We added a word ("written") in the section ethics approval and consent to participate as well as a phrase ("Both of them gave the permission to publish their names.") in the acknowledgements.

Kind regards,

Stefan Gysin